Author's response to reviews

Title: Pudendal nerve decompression in perineology: a case series.

Authors:

Jacques Beco (jacques.beco@skynet.be)
Daniela Climov (dclimov@parnasse-deuxalice.edu)
Michele Bex (michele.bex@skynet.be)

Version: 10 Date: 26 October 2004

Author's response to reviews:

"Figure 2 can be published thanks to the patient's consent." has been introduced in the acknowledgements section. The number 74 has been replace in the right position in table 5.